Author's response to reviews

Title: Evaluation of effectiveness of class-based nutrition intervention on changes in soft drink and milk consumption among young adults.

Authors:

Eun-Jeong Ha (eha@kent.edu)
Natalie Caine-Bish (ncaine@kent.edu)
Christopher Holloman (holloman@stat.osu.edu)
Karen Lowry-Gordon (klowry@kent.edu)

Version: 2 Date: 3 July 2009

Author's response to reviews:

July 3, 2009

Dear Editor,

Thank you for the reviewer and editor's comments for the manuscript entitled “Evaluation of effectiveness of class-based nutrition intervention on changes in soft drink and milk consumption among young adults”. We were pleased to know that our manuscript was rated as potentially acceptable for publication in Nutrition Journal, subject to adequate revision and response to the comments raised by the reviewers.

As you notice, we have revised the manuscript based on the comments made by the reviewers. Attached is our point-by-point response to the comments raised by the reviewers. We would like to express our thanks to the reviewers who identified areas of our manuscript that needed corrections or modification. If you need anything further, please don't hesitate to contact me at 330-672-2701 or eha@kent.edu

Sincerely Yours,
Eun-Jeong Ha, Ph.D.
Assistant Professor
Nutrition and Dietetics